Author's response to reviews

Title: Randomised controlled trial of cervical radiofrequency lesions as a treatment for cervicogenic headache [ISRCTN 07444684]

Authors:

Sarah RS Haspeslagh (shas@anaesth.azm.nl)
Hans A van Suijilekom (j.a.van.suijilekom@freeler.nl)
Inge E Lame (iel@sane.azm.nl)
Alfons GH Kessels (akes@kemta.azm.nl)
Maarten van Kleef (mvk@sane.azm.nl)
Wim EJ Weber (wweb@neurologie.azm.nl)

Version: 3 Date: 5 December 2005

Author's response to reviews:

In the abstract we have added transcutaneous electrical nerve stimulation and quality of life for TENS and RAND-36 (RAND means research and development, and is, as such, not used in its full name; RAND-36 is a well-known Quality of life measurement).

We have Background substituted for Introduction.

We have altered the order of the manuscript, as asked.

We have deleted the figures from the manuscript and uploaded them separately.

Wim Weber